

# Advances in Mechanics of Soft Materials: A Review of Large Deformation Behavior of Hydrogels

Zishun Liu\*

International Center for Applied Mechanics State Key Laboratory for Strength and Vibration of Mechanical Structures, Xi'an Jiaotong University Xi'an 710049, P. R. China zishunliu@mail.xjtu.edu.cn

William  $\mathrm{Toh}^\dagger$  and Teng Yong  $\mathrm{Ng}^\ddagger$ 

School of Mechanical and Aerospace Engineering Nanyang Technological University 50 Nanyang Avenue, Singapore 639798, Singapore <sup>†</sup>wtoh1@e.ntu.edu.sg <sup>‡</sup>mtyng@ntu.edu.sg

> Received 6 August 2015 Revised 11 August 2015 Accepted 11 August 2015 Published 19 October 2015

Hydrogels possess magnificent properties which may be harnessed for novel applications. However, this is not achievable if the mechanical behaviors of hydrogels are not well understood. This paper aims to provide the reader with a bird's eye view of the mechanics of hydrogels, in particular the theories associated with deformation of hydrogels, the phenomena that are commonly observed, and recent developments in applications of hydrogels. Besides theoretical analyses and experimental observations, another feature of this paper is to provide an overview of how mechanics can be applied.

Keywords: Hydrogel; large deformation; mechanical behavior; thermodynamic theory.

# 1. Introduction

When a solvent comes into contact with a network of hydrophilic crosslinked polymer chains, the attraction between the two species causes solvent molecules to be entrapped within the network, thus causing a change in volume due to the imbibement of solvent molecules. The resulting swollen state is commonly known as a hydrogel. Possessing superior capabilities to imbibe solvents and swell to a large

\*Corresponding author.

extent, hydrophilic polymers, or hydrogels, are an interesting class of materials which present tremendous potential as the next generation material of choice.

Being biocompatible, many applications have been developed for hydrogels in the past few decades. Early uses of hydrogels tap on its superior bio-compatibility over plastic materials [Wichterle and Lim, 1960]. Some examples include contact lenses, wound dressings and implants [Corkhill *et al.*, 1989; Kuroyanagi, 1999; Wichterle and Lim, 1960]. Consisting of sparse crosslinks in the hydrophilic polymer network, gels are able to imbibe and retain water, causing a swelling of as large as 1000 times its initial volume [Oguz, 2007].

With different monomers or constituent particles, hydrogels are able to undergo large deformation with a small change in environmental stimulus, but not limited to temperature, pH value, light, electric field, ionic strength and magnetic field [Aguilar et al., 2007; Jeong and Gutowska, 2002; Li and Kong, 2007; Meng and Hu, 2010]. These environmentally sensitive gels are also known as "smart hydrogels" and are more attractive than the traditional hydrogels, whose main attraction is its superabsorbency. However, the scope of traditional hydrogel applications is often severely limited by their mechanical behavior. For example, most hydrogels do not exhibit sufficiently high stretchability. Gong' group from Hokkaido University have successfully developed the highly strong, tough, and viscous hydrogels which have a hierarchically-ordered complex structure. These types of hydrogels will provide revolutionary applications over traditional hydrogels [Gong, 2010; Gong et al., 2003; Sun et al., 2013]. Their recent inventions of tough hydrogels demonstrated the potential as structural materials. These physical hydrogels have a combination of mechanical properties including stiffness, strength, toughness, fatigue resistance and self-healing, along with biocompatibility. Most recently, the Suo group from Harvard University has developed a new method for the synthesis of hydrogels from polymers forming ionically and covalently crosslinked networks [Keplinger *et al.*, 2013. These types of gels can serve as model systems to explore mechanisms of deformation and energy dissipation, and expand the scope of hydrogel applications.

Recently, there has been a shift in interests for the choice of materials used in machines. Traditionally, machines are associated with strong and hard engineering materials, such as metals and ceramics, which are very limited by the small deformation that they are able to undergo. With more research into smart hydrogels, researchers have been attracted to tap into the new realm of soft machines, which tend to mimic what we observe in nature. The potential applications involving hydrogels has grown exponentially to include areas such as biomedical systems, actuators and flow control [Calvert, 2009; Satarkar *et al.*, 2010; Tomatsu*et al.*, 2011; Ward and Georgiou, 2011].

With the fast expanding potential for more applications in a diverse array of areas, a better understanding of the mechanics of hydrogels is imperative. In recent times, many theories have been developed to describe and understand the swelling behavior of hydrogels under the influence of various environmental stimuli and mechanical constraints. With a better understanding of the mechanics of hydrogels, it is possible to design more applications with improved performance.

In this paper, we review some of the recent works aligned with the direction of providing a better understanding of gel mechanics, including earlier multiphase theories and more recently, the monophase theories, spearheaded by the Suo Group. Following the description of these theories, the paper then proceeds to discuss some phenomena commonly observed in gels, and how mechanics is used to study these phenomena.

The paper is organized as follows. Section 2 provides some theories on the equilibrium swelling of a gel. Section 3 covers the kinetics of swelling. Section 4 looks at the efforts in developing modeling and simulation tools. Section 5 discusses the phase transition phenomenon present in gels. Section 6 investigates the intriguing phenomenon of swelling induced instabilities.

# 2. Large Deformation Theories of Hydrogels

Under the influence of external stimuli, a hydrogel undergoes deformation from its initial reference state to the final current state. In describing this state of deformation, the deformation gradient  $F_{iK}$  (or **F**) is conventionally used. Figure 1 shows the state of deformation of a hydrogel.

It is defined as the partial derivative of the current state of the gel,  $x_i$ , with respect to the reference state  $X_K$ , i.e.,

$$F_{iK}(\mathbf{X},t) = \frac{\partial x_i(\mathbf{X},t)}{\partial X_K}.$$
(1)

In equilibrium, the change in free energy of the gel is balanced by the external work done on the gel. This thermodynamic equilibrium is usually written in the



Fig. 1. A hydrogel undergoing deformation under the effects of external stimuli and changes from the reference state  $X_K$  to the current state  $x_i$ .

form

$$\int_{V} \delta W dV = \int_{V} B_{i} \delta x_{i} dV + \int_{A} T_{i} \delta x_{i} dA + \sum \left( \mu^{r} \int_{V} \delta C^{r} dV \right), \tag{2}$$

where W is the free energy of the gel;  $B_i$  is the external body force;  $T_i$  is traction;  $\mu^r$  and  $C^r$  the chemical potential and concentration of the r species, which includes solvent (s) for all gels, anion (-) and cations (+) for pH sensitive gels, photochemical reactions (p) for photo-thermal gels.

By equating the change in free energy in the gel to external work done on the gel, the nominal stress  $s_{iK}$  is shown to be the partial derivative of the free energy function with respect to the deformation gradient, i.e.,

$$s_{iK} = \frac{\partial W}{\partial F_{iK}}.$$
(3)

The chemical potential of solvent within the gel is shown to be

$$\mu^s = \frac{\partial W}{\partial C^s}.\tag{4}$$

# 2.1. Free energy functions

The history of hydrogel deformation theory begins with pioneering works by Flory and Rehner [1943], suggesting that the Helmholtz free energy of a polymer network in an aqueous solution can be written as the additive decomposition of the free energy of elastic stretch of the polymer network and the free energy of mixing when the polymer network interacts with a solution. This decomposition has been widely used in the theories of hydrogels. According to Flory theory, the more general form of free energy of hydrogel can be described as

$$W = W_{\text{net}} + W_{\text{mix}} + \sum_{r \neq s} W_r, \qquad (5)$$

where  $W_{\text{net}}$  and  $W_{\text{mix}}$  are the free energy of network stretch and free energy of mixing, respectively. These two entities are applicable to all types of hydrogels.  $W_r$  represents the free energy contributions due to factors other than solvent, which are specific to certain types of gels.

These free energies include, but are not limited to, the free energy of ionization  $W_{\rm ion}$ , polarization  $W_{\rm pol}$  and dissociation  $W_{\rm dis}$ , commonly found in polyelectrolyte or pH-sensitive hydrogels; free energy of photo-chemical reactions  $W_{\rm pho}$  in photo-thermal sensitive hydrogels; free energy of magnetization  $W_{\rm mag}$  in ferrogels. In addition, the Landau free energy of phase transformation  $W_{\rm phase}$  is used to represent contributions from the phase transformation process in gels. The explicit forms of each of these entities are described in the following sub-sections.

#### 2.1.1. Free energy of network stretch

In all hydrogels, a deformation leads to a change in the polymer network, resulting in a change of the free energy. This contributes to the free energy of the gel in the form of the free energy of network stretch,  $W_{\text{net}}$ .

Without consideration for solvent action, the free energy of elastic stretch considers solely the stretch of the dry polymer network and is akin to the stretch of a rubber. Many forms for this free energy of rubber hyperelasticity exists [Anand, 1996; Arruda and Boyce, 1993; Boyce and Arruda, 2000; Flory and Rehner, 1943; Lopez-Pamies, 2010; Marckmann and Verron, 2006; Valanis and Landel, 1967]. The most commonly used rubber hyperelasticity model used in hydrogels is the Flory model [Flory, 1953], based on Gaussian statistics.

$$W_{\rm net}(\mathbf{F}) = \frac{1}{2} NkT[\operatorname{tr}(\mathbf{F}^T \mathbf{F}) - 3 - 2\ln(\det \mathbf{F})], \qquad (6)$$

where N is the number of effective chains in the network and kT, product of Boltzmann constant k and temperature T, is a measure of temperature in the units of energy.

It has been noted that models lacking consideration of limited extensibility of the polymer network [Chester and Anand, 2010; Chester *et al.*, 2014] and non-Gaussian statistical models [Anand, 1996; Arruda and Boyce, 1993] have been used alternatively [Chester, 2012; Chester and Anand, 2010, 2011; Chester *et al.*, 2014] in the form

$$W_{\rm net}(\mathbf{F}) = -NkT \left\{ -\lambda_L^2 \left[ \left( \frac{\bar{\lambda}}{\lambda_L} \right) \beta + \ln \left( \frac{\beta}{\sinh \beta} \right) - \left( \frac{1}{\lambda_L} \right) \beta_0 - \ln \left( \frac{\beta_0}{\sinh \beta_0} \right) \right] + \left( \frac{\lambda_L}{3} \beta_0 \right) \ln(\det \mathbf{F}) \right\},$$
(7)

where  $\bar{\lambda} = \sqrt{\operatorname{tr}(\mathbf{F}^T \mathbf{F})/3}$ ,  $\lambda_L$  is the network locking stretch parameter,  $\beta = L^{-1}(\bar{\lambda}/\lambda_L)$ ,  $\beta_0 = L^{-1}(1/\lambda_L)$ ,  $L^{-1}$  is the inverse of the Langevin function  $L(x) = \operatorname{coth} x - x^{-1}$ .

#### 2.1.2. Free energy of mixing

As a hydrogel deforms, the volumetric change is due to the absorption or desorption of solvent molecules. The interaction between the polymer network and solvent contributes to the free energy of the system in the form of free energy of mixing.

The free energy of mixing is most commonly associated with the Flory–Huggins model [Flory, 1942; Huggins, 1941], given by the form

$$W_m(C^s) = -\frac{kT}{\nu} \left[ \nu^s C^s \ln\left(\frac{1+\nu^s C^s}{\nu^s C^s}\right) + \frac{\chi \nu^s C^s}{1+\nu^s C^s} \right],\tag{8}$$

where  $C^s$  is the concentration of solvent molecules in the reference state,  $\nu^s$  is the volume per small molecule and  $\chi$  is the Flory interaction parameter, measuring the extent of mixing between solvent and polymer network.

In the isothermal analysis of gels, the Flory interaction parameter is usually taken to be a constant in the range of  $0 < \chi < 1.2$ . In temperature sensitive hydrogels, the interaction parameter is dependent on temperature and polymer concentration, originally given in the form [Huggins, 1964]

$$\chi(\phi, T) = \chi_h(\phi) + \chi_s(\phi, T), \tag{9}$$

representing the enthalpic and entropic contributions, respectively. Various experimental fittings have been performed for the dependence of  $\chi$  on temperature and concentration [Afroze *et al.*, 2000; Caykara *et al.*, 2006; Erman and Flory, 1986; Hassan and Durning, 1999; Kojima *et al.*, 2013; Oliveira *et al.*, 2004; Shimizu *et al.*, 2003; Shirota *et al.*, 1998] and adopted in thermodynamic theories for temperature sensitive hydrogel models [Birgersson *et al.*, 2008; Cai and Suo, 2011; Hino and Prausnitz, 1998; Hu *et al.*, 2013a; Li *et al.*, 2005b].

#### 2.1.3. Free energy of ionization

For polyeletrolyte gels, the charges in the ions give rise to ionic interactions, which include ionization and polarization. The ionization process contributes to a change in the free energy [Brannon-Peppas and Peppas, 1991; Hong *et al.*, 2010; Marcombe *et al.*, 2010; Ricka and Tanaka, 1984; Yan *et al.*, 2014].

The free energy of ionization is assumed to be entirely due to the entropy of mixing results from low concentration of ions and is given as the sum of all species r of mobile ions except the solvent,

$$W_{\rm ion} = kT \sum_{r \neq s} C^r \left( \ln \frac{C^r}{\nu^s C^s c_0^r} - 1 \right),\tag{10}$$

where  $C^r$  is the nominal concentration of species r and  $c_0^r$  the reference concentration of species r in the solution.

# 2.1.4. Free energy of polarization

Due to uneven distribution of charges within the gel, the free energy is altered by the contribution of the free energy of polarization [Hong*et al.*, 2010; Yan *et al.*, 2014], given by the expression

$$W_{\rm pol} = \frac{1}{2\varepsilon} \frac{F_{iK} F_{iL}}{\det \mathbf{F}} \tilde{D}_K \tilde{D}_L, \qquad (11)$$

where  $\varepsilon$  is the permittivity of the gel and  $\tilde{D}_i$  is the nominal electric displacement.

#### 1530001-6

# 2.1.5. Free energy of dissociation

In solutions consisting of weak acids, there is partial dissociation of the acidic group AH into the hydrogen ions  $H^+$  and conjugate base  $A^-$ 

$$AH \leftrightarrow A^- + H^+. \tag{12}$$

This dissociation contributes to the free energy of the system [Marcombe *et al.*, 2010; Ricka and Tanaka, 1984] and is given by

$$W_{\rm dis} = kT \left[ C_{A^-} \ln \left( \frac{C^{A^-}}{C^{A^-} + C^{AH}} \right) + C^{AH} \ln \left( \frac{C^{AH}}{C^{A^-} + C^{AH}} \right) \right] + \gamma C^{A^-}, \quad (13)$$

where  $C^r$  are the nominal concentrations of the *r*th species of ions and  $\gamma$  is the increase in enthalpy when an acid dissociates.

# 2.1.6. Free energy of photo-excitation

In a photo-thermal gel, the light absorbing nanoparticles absorb light of particular frequencies f. This absorption results in a spontaneous excitation and decay of the outer electron, which may be represented by the chemical equation

$$X + \hbar f \to Y, \quad Y \to X + \text{heat},$$
 (14)

where X is the ground state particle, and Y the excited state and  $\hbar f$  the energy possessed in a single photon of frequency f.

The free energy molecular excitation of the nanoparticles is associated with the product of number of photo-chemical reactions  $C^p$  and the affinity associated with a single photo-chemical reaction [Parson, 1978]

$$W_{\rm photo} = C^p \left[ \hbar f + kT \ln \left( \frac{C^p}{C^{g,0}} \right) \right]. \tag{15}$$

#### 2.1.7. Free energy of magnetization

When a gel is infused with magnetic particles and exposed to a magnetic field, the magnetization of the particles have a contribution to the free energy of the gel [Han *et al.*, 2011]. One form of this free energy used is given by

$$W_{\rm mag} = \frac{1}{2\mu^B} \mathbf{B} \cdot \mathbf{B},\tag{16}$$

where  $\mu^B$  is the magnetic permeability and **B** the magnetic induction. Equation (16) is a highly simplified expression of the free energy as it assumes no hysteresis and linear magnetic properties.

#### 1530001-7

# 2.1.8. Free energy of phase transformation

In gels which undergo phase transformation, the Landau theory for phase transformation is sometimes used [Drozdov, 2014a, 2014b]. It assumes a free energy contribution of the form

$$W_{\text{phase}} = \frac{1}{\nu} \left[ \frac{1}{2} A_1 (T - T_*) \eta^2 + \frac{1}{4} A_2 \eta^4 + \frac{1}{6} A_3 \eta^6 \right], \tag{17}$$

where  $A_1$ ,  $A_2$  and  $A_3$  are constant coefficients,  $T_*$  the phase transition temperature and  $\eta$  is the nematic order parameter.

#### 2.2. Physical constraints for different hydrogels

Physically, the constituents of the gel have to follow conservation laws. Some common physical constraints which have been used in the modeling of various types of gels are given in the following sections.

#### 2.2.1. Molecular incompressibility constraint

During deformation, a gel changes volume by absorption and desorption of solvent molecules across the boundary. There is negligible change in the volumes of individual molecules as the stress that is present during swelling is insufficient to cause deformation at the molecular level.

Under this assumption, it is safe to assume that the volume of the gel is equal to the sum of volume of the constituent polymer and the solvent within the network. This constraint is most commonly expressed as

$$1 + \nu C^s = \det \mathbf{F}.\tag{18}$$

#### 2.2.2. Electroneutrality

As a simplification in polyelectrolyte gels, it is assumed that the overall charge on the gel and in the external solution is neutral. This assumption gives rise to the following expressions

$$C^{H^+} + C^+ = C^{A^-} + C^-, (19)$$

$$\bar{n}^{H^+} + \bar{n}^+ = \bar{n}^-,\tag{20}$$

where  $C^r$  are the nominal concentrations of the superscripted species and  $\bar{n}^r$  the number of particles of the superscripted species in the external solution.

#### 2.2.3. Energy conservation

In photothermal sensitive hydrogels, the photo-excitation of light absorbing particles is a spontaneous process. The energy absorbed from light excites an electron to the outer shell and is immediately demoted back to its original energy level, and the energy released from this demotion is converted into heat energy.

Due to near unity conversion from light energy to thermal energy, energy conversion can be written in the form

$$[c_v^{\text{network}}\phi + c_v^s(1-\phi)]\alpha I_0 = hfC^p \det \mathbf{F},$$
(21)

where the left-hand side corresponds to thermal energy and the right-hand side corresponds to light energy.  $c_v^r$  represents the volumetric heat capacity of the network and solvent molecules,  $\alpha$  is a proportionality constant,  $I_0$  is the intensity of the monochromatic light irradiation with frequency f,  $\hbar$  is the Planck constant, and  $C^p$ the nominal concentration of photo-chemical reactions taking place.

# 2.3. Imposition of physical constraints

#### 2.3.1. Lagrange multiplier

The enforcement of these physical constrains are usually carried out in two ways. First through the addition of Lagrange multipliers to the thermodynamic equilibrium equation [Baek and Pence, 2011; Baek and Srinivasa, 2004; Duda *et al.*, 2010; Hong *et al.*, 2010; Hong *et al.*, 2008b]. This method leads to nominal stress and chemical potentials being functions of the Lagrange multiplier, as seen in Eqs. (22) and (23) for the case of a neutral gel

$$s_{iK} = \frac{\partial W}{\partial F_{iK}} - \Pi H_{iK} \det \mathbf{F}, \qquad (22)$$

$$\mu^s = \frac{\partial W}{\partial C^s} + \Pi \nu^s. \tag{23}$$

The Lagrange multiplier  $\Pi$  is an inhomogeneous field, solvable as part of initial and boundary value problems. In a neutral hydrogel, it is analogous to the osmotic pressure due to the external solvent. Li *et al.* [2012] and Cai and Suo [2012] investigated  $\Pi$  as an equation of state for ideal elastomeric gels.

# 2.3.2. Legendre transformation

An alternative method is through the use of a Legendre transformation of the free energy function [Dinget al., 2013; Hong et al., 2009a; Marcombe et al., 2010; Toh et al., 2014a]

$$\hat{W} = W - \sum \mu^r C^r, \tag{24}$$

where expressions for  $C^r$  are obtained using the constraint equations (18), (19) and (21).

The transformation converts the thermodynamic equilibrium equation into one which is represented by a hyperelastic solid

$$\int_{V} \delta \hat{W} dV = \int_{V} B_i \delta x_i dV + \int_{A} T_i \delta x_i dA.$$
<sup>(25)</sup>

Under this definition, the nominal stress is shown to be

$$s_{iK} = \frac{\partial W}{\partial F_{iK}}.$$
(26)

The constitutive equation of gel materials may be defined using the forth order tangent modulus tensor,  $C_{ijkl}$  [Kang and Huang, 2010c].

In equilibrium, the concentration of solvent within the swollen gel is homogeneous in the absence of external mechanical constraints. However, in the presence of such constraints, the concentration becomes inhomogeneous despite being in mechanical equilibrium [Hong *et al.*, 2009a; Pritchard and Terentjev, 2013; Zhao *et al.*, 2008]. In addition, the presence of a surface causes a shift in equilibrium of the gel between the saturated state, unsaturated state and a gel in equilibrium with a vapor of its own liquid [Baek and Pence, 2011].

A widely observed phenomenon in gels is volume phase transition, where a small change in stimulus causes a sudden large deformation in the gel. Experimental results have shown that the phase transition process can be continuous or discontinuous at different times, depending on the constituents and environmental conditions [Suzuki, 1993]. Factors influencing this shift from continuous to discontinuous phase transition include, but are not limited to, light, temperature, applied stress and pH-value [Kondo *et al.*, 1993; Suzuki *et al.*, 1996; Suzuki and Kojima, 1994; Suzuki and Suzuki, 1995; Suzuki and Tanaka, 1990]. Various models have been proposed to study this phenomenon [Cai and Suo, 2011; Toh *et al.*, 2014a; Wang, 2007]. The Landau free energy function for phase transition has also been used to circumvent the discontinuity in phase transitions [Drozdov, 2014a, 2014b].

# 3. Swelling Kinetics of Hydrogels

#### 3.1. Multi-phase theories

Earlier theories of gel swelling were based on considering the polymer network and solvent as two separate phases. Works by Tanaka *et al.* made use of equations derived based on Newton's second law with inclusion of the effects of friction, relative motion and viscosity between the phases [Peters and Candau, 1986, 1988; Tanaka and Fillmore, 1979; Tanaka *et al.*, 1973]. Later, a two-step mechanism was proposed to generalize the swelling kinetics of gels of arbitrary shapes [Li and Tanaka, 1990].

A coupling relation between stress and gel composition was proposed by using phenomenological hydrodynamic relations, treating the gel as a two-fluid model [Calderer *et al.*, 2008; Doi, 2009; Doi and Onuki, 1992; Yamaue and Doi, 2004a; Yamaue and Doi, 2004b; Yamaue and Doi, 2005].

These theories provide a very detailed analysis of gel properties. However, they are limited by the relative difficulty in implementing simulations to aid researchers in understanding material response under more complex situations.

# 3.2. Mono-phase theory of hydrogel

The swelling of a gel using a continuum approach was pioneered by Durning and Morman [1993] to overcome the shortcomings of the hydrodynamic models proposed in the multi-phase approach, motivated by the relative complexity of multi-phase theories and also the ease of single continuum implementation. Back and Srinivasa [2004] proposed a monophase theory for a solvent diffusing through an elastic solid.

However, monophase theories were not favored until more recently, when the Suo Group from Harvard proposed a more unified thermodynamic framework coupling diffusion and large deformation in a gel [Hong *et al.*, 2008b]. With many promising applications lying ahead for hydrogels, mono-phase theories have gained increasing popularity against multi-phase theories due to the relative ease of modeling and simulation of material behaviors. This thermodynamic framework has spurred much interests in the monophase theory, as seen from the numerous works stemming from the same framework [An *et al.*, 2010; Bouklas and Huang, 2012; Chester, 2012; Chester and Anand, 2010, 2011; Duda *et al.*, 2010; Hong *et al.*, 2010; Lucantonio *et al.*, 2013].

Whilst using the Flory–Rehner model has been widely adopted for study of gel swelling, there lies a less than unified theory for the study of gel swelling kinetics, as some theories consider the polymer network and fluid to be distinct phases for the robustness. Some theories treat the polymer-fluid interaction as a single phase by coupling the diffusion of solvent with the large deformation of the network, simplifying the development process of numerical simulation methods used to predict the response of hydrogels.

The deformation gradient is often based on a multiplicative decomposition of elastic and swelling parts [Boyce and Arruda, 2001; Chester and Anand, 2010; Duda *et al.*, 2010], written as

$$\mathbf{F} = \mathbf{F}^e \mathbf{F}^s. \tag{27}$$

However, it was noted that in the swelling of a polymeric gel over an extended time-span, the elastic part of the deformation may be taken to be an instantaneous process and for modeling of the migration of solvent molecules into the gel, it may be taken to be unity [Hong *et al.*, 2008b]. Indeed, in equilibrium studies, we are concerned with the end state deformation rather than the initial change in deformation.

# 3.3. Neutral gel

Baek and Srinivasa [2004], Hong *et al.* [2008b], Duda *et al.* [2010] and Chester and Anand [2010] have developed thermodynamic theories which couple diffusion and large deformation, paving the way for future works in the continuum mechanics studies of hydrogel swelling. The governing equations of gel swelling are dependent on the migration of solvent molecules. The flux is given by

$$j_i = \frac{cD}{kT} \frac{\partial \mu}{\partial x_i},\tag{28}$$

and it follows conservation laws, given by the form

$$\int_{V} \frac{1}{\det \mathbf{F}} \frac{\partial C}{\partial t} dV + \int_{A} j_{i} dA = 0.$$
<sup>(29)</sup>

Based on the coupled diffusion-deformation theory, Yoon *et al.* [2010] developed a linearized poroelastic swelling kinetic theory to characterize material properties such as the modulus, Poisson's ratio and permeability of a gel swelling within the regime of linear poroelasticity. A comparison with the nonlinear theory has shown the linear theory to be consistent within the linear portion of swelling [Bouklas and Huang, 2012].

#### 3.4. Temperature sensitive gel

Birgersson *et al.* [2008] developed a model for the transient response of a temperature sensitive hydrogel by considering conservation laws for the polymer and fluid phase separately using the mixture theory. Analysis takes place in a nondimensional framework to elucidate important characteristics as well as simplify the governing equations. In addition to fluid transport, the kinetic laws of a temperature sensitive hydrogel also depends on the temperature gradient of the gel, which may be modeled using Fourier's laws of heat conduction, given in Eqs. (30) and (31)

$$q_i = -\kappa \frac{\partial T}{\partial x_i},\tag{30}$$

$$\int_{V} \rho \frac{dU}{dt} dV + \int_{A} q_{i} dA = \int_{V} r dV.$$
(31)

By considering entropy imbalance, Chester and Anand [2011] extended the coupled theory for fluid permeation in elastomeric materials [Chester and Anand, 2010] to account for thermal response in temperature-sensitive hydrogels.

# 3.5. Polyelectrolyte gel

Kinetic theory of polyelectrolyte gels (pH gels) are mostly assumed to follow the Nernst–Planck equation of ion diffusion kinetics [Helfferich and Plesset, 1958]. Other than kinetic laws based on diffusion, the distribution of ions within the gel is also dependent on ionic interactions, following the Nernst–Planck equation

$$j_i^r = -D^r \left[ \frac{\partial C^r}{\partial x_i} + z^r C^r \left( \frac{\mathbf{F}}{RT} \right) \frac{\partial \psi}{\partial x_i} \right], \tag{32}$$

where the superscript r denote the rth species of ion,  $D^r$  is the diffusion coefficient,  $C^r$  is the concentration,  $z^r$  is the valency number,  $\mathbf{F}$  is the Faraday

constant, R is the gas constant, T is the temperature, and  $\psi$  is the electric potential.

# 4. Numerical Simulation of Hydrogels

To predict the behavior of hydrogels, numerical simulation is commonly used as it allows for studying gels of complex geometries which are often too complicated for any analytical solutions.

Common simulation tools used in the study of hydrogel behaviors include finite element method, meshless methods and molecular dynamics simulations.

# 4.1. Finite element method

With the development of monophase theories, finite element simulation of hydrogel swelling has been expedited, in particular with the use of finite element software ABAQUS for its versatility in defining material models through user-defined subroutines. Besides ABAQUS, there has also been finite element simulations performed using COMSOL Multiphysics and on other platforms. Nevertheless, it should be noted that simulation using ABAQUS is one of the most popular methods as the solver is capable of solving highly nonlinear cases. Common subroutines used for equilibrium swelling include UHYPER [Ding *et al.*, 2013; Hong *et al.*, 2009a; Marcombe *et al.*, 2010; Toh *et al.*, 2014a], which models the equilibrium swelling of a gel as a hyperelastic material through a Legendre transformation; and UMAT [Ding *et al.*, Under review; Kang and Huang, 2010c], which requires the tangent modulus tensor as input. Advantages of UHYPER over UMAT include easy implementation and less computational time. On the other hand, UMAT provides a more robust simulation, enabling the use of anisotropic initial conditions, as UHYPER requires isotropic initial conditions.

The finite element models for swelling kinetics of gels include the use of one or more subroutines used concurrently. Zhang *et al.* [2009] and Chester *et al.* [2012; 2011; 2014] have made substantial contributions in the area of developing userdefined elements (UEL subroutine) for the simulation of neutral, viscoelastic and temperature sensitive hydrogels. It should be noted that the development of UEL subroutines is a tedious process, as it involves much work on the discretization of elements. A discretization has to be performed for every element type that is required for simulation. Less rigorous methods have also been developed for the same purpose by utilizing the coupled temperature-displacement elements already present in the ABAQUS library, thus eliminating the need for re-formulation of elements [Duan *et al.*, 2013; Toh *et al.*, 2013; Toh *et al.*, 2014b].

Owing to the powerful multiphysics coupling mechanism present, COMSOL Multiphysics is a popular software of choice for simulation of heavily coupled physical material models. The software has been used to simulate transient swelling kinetics of polymeric gels [Lucantonio *et al.*, 2013], viscoelastic polymer gels [Li *et al.*, 2013b; Wang and Hong, 2012], temperature sensitive gels [Birgersson *et al.*, 2008] and ferrogels [Han *et al.*, 2011].

Wallmersperger *et al.* [2011a, 2011b] developed a chemo-electro-mechanical finite element model to solve for the equilibrium swelling of pH-sensitive gels, taking into account effects of various ions and their interactions. Liu *et al.* [2015] developed a finite element algorithm based on the multiplicative decomposition of deformation gradient to simulate the inhomogeneous swelling of neutral hydrogels in equilibrium state.

Dolbow *et al.* studied the kinetics of hydrogels based on the motion of the sharp interface between difference phases of the gel using the extended FEM (XFEM) method [Dolbow *et al.*, 2004, 2005; Ji *et al.*, 2006].

# 4.2. Meshless methods

Apart from the finite element method, meshless methods are also very popular for simulation of hydrogels. However, for meshless methods, there is a wide spectrum of approaches which has been used in the modeling of gel materials. These include the improved complex variable element-free Galerkin (ICVEFG) method on neutral hydrogels [Li *et al.*, 2014], strong-form meshless random differential quadrature (RDQ) method [Li and Mulay, 2011], and finite cloud method on pH-sensitive gels [De and Aluru, 2004; De *et al.*, 2002] and Hermite–Cloud method on electricsensitive hydrogels [Chen *et al.*, 2005; Lam *et al.*, 2006; Li, 2009; Li *et al.*, 2003, 2006, 2007a; Li *et al.*, 2007c; Luo *et al.*, 2007]; pH-sensitive hydrogels [Li *et al.*, 2005a; Li and Yew, 2009; Li *et al.*, 2004; Li *et al.*, 2005c; Ng *et al.*, 2010]; electric-pH sensitive hydrogels [Li *et al.*, 2007b; Luo *et al.*, 2008]; and ionic strength sensitive hydrogels [Lai and Li, 2010, 2011; Li and Lai, 2011].

#### 4.3. Molecular dynamics simulations

In simulations of smaller length scales, variations of molecular dynamics (MD) has also been used, such as a combined discontinuous molecular dynamics and Monte Carlo technique [Kenkare *et al.*, 2000], as well as coarse grained molecular dynamics [Quesada–Perez *et al.*, 2012]. In coarse grained dynamics model, Quesada–Perez *et al.* adopted coarse grained molecular dynamics to simulate the steady state swelling behavior of gels due to temperature and pH changes. The truncated Leonard–Jones potential was used for interaction between particles and a hydrophobic interaction potential. Jaramillo–Botero *et al.* [2010] investigated the thermodynamic and mechanical properties of polymer-based hydrogel networks by using atomistic-level molecular dynamics. The study provided insights into elastic response, cohesive energies, viscosities, and stress–strain relationships for relevant single and double network hydrogel compositions from atomistic level. The works can be used for steering experimental efforts toward systematic optimization of the bio-mimetic response of polymer-based scaffolds in tissue engineering. Sliozberg

et al. [2014] used computational coarse-grained model to design complex polymer networks and gels, in which the enhanced and tunable mechanical properties are obtained. The elastic properties and deformation mechanisms of the polymers were also investigated by tensile test at various strain rates. Their studies demonstrated that the architecture of the polymer can be optimized and thus the elastic properties can be tuned for specific engineering applications.

# 5. Some Phenomenon Related to Gel Deformation

# 5.1. Phase transition

In the gel deformation process, it is commonly observed that across a critical loading parameter, the volume change of the gels occurs drastically. This phenomenon is known as the phase transition of a gel. Experimentally, the phase transition has been observed in hydrogels responsive to different types of environmental stimuli. The Tanaka group from MIT has performed substantial work on the phase transitions in various types of gels, which include ionic gels [Ohmine and Tanaka, 1982; Tanaka *et al.*, 1980], nonionic gels [Amiya *et al.*, 1987; Hirotsu *et al.*, 1987] and photosensitive gels [Mamada *et al.*, 1990; Suzuki and Tanaka, 1990]. Suzuki *et al.* from the Yokohama National University of Japan explored the effects of several factors on the phase transition of gels, including uniaxial stress [Kondo *et al.*, 1993; Suzuki and Kojima, 1994], pH-value [Suzuki and Suzuki, 1995] and light [Suzuki and Tanaka, 1990]. Juodkazis *et al.* [2000] studied the effects of radiation forces on hydrogels.

Cai and Suo [2011] developed a thermodynamic theory of deformation in temperature sensitive hydrogels and studied the phase transition phenomenon of temperature sensitive hydrogels. In addition, the model was applied to study the phase coexistence between gels of different phases. Ding *et al.* [2013] reported that in temperature sensitive hydrogels, other than the critical temperature, there is also a critical chemical potential which causes phase transition. Toh *et al.* [2014a] developed a thermodynamic theory for photo-thermal sensitive hydrogels and studied the effects of light intensity on the phase transition.

# 5.2. Instabilities

Mechanical instabilities, such as wrinkling, creasing, buckling and bifurcation are commonly observed phenomenon in hydrogels [Barros*et al.*, 2012; Douezan *et al.*, 2011; DuPont Jr. *et al.*, 2010; Guvendiren *et al.*, 2009; Ji and Ding, 2002; Kim *et al.*, 2010; Klein *et al.*, 2007; Lee *et al.*, 2014; Liang and Mahadevan, 2011; Matsuo and Tanaka, 1992; Peixinho and Mukhopadhyay, 2013; Savin *et al.*, 2011; Sultan and Boudaoud, 2008; Tanaka *et al.*, 1992; Tanaka *et al.*, 1987; Wu *et al.*, 2014; Zhu *et al.*, 2012]. These mechanical instabilities, initially deemed to be undesirable, have recently become the subject of interest, with the instabilities exploited for useful applications, such as tunable adhesion, micro-patterning, particle sorting, microfluidics, microfabrication and actuation [Chen and Yang, 2012; Chen and Yin, 2010; Hu *et al.*, 1998; Kwon *et al.*, 2010; Yang *et al.*, 2010].

These exciting potential applications have spurred many theoretical analyses of the instability, with analyses in several broad areas, including:

#### 5.2.1. Bilayer structures

The most commonly studied model for swelling induced instabilities in soft materials is the bilayer structure, where a thin layer of gel film is bounded to a substrate (Fig. 2), such as an elastic substrate [Cai *et al.*, 2011; Huang and Suo, 2002a, 2002b; Huang *et al.*, 2004; Jia and Ben Amar, 2013; Kang and Huang, 2010b; Liu *et al.*, 2010; Liu *et al.*, 2011] or a gel substrate of different material properties [Wu *et al.*, 2013].

The buckling of bilayer structures present many interesting buckling patterns, such as the herringbone, labyrinth patterns, peanut, lamellar and hexagonal structures [Cai *et al.*, 2011; Guvendiren *et al.*, 2009].

For soft-on-hard bilayer structures, creasing instability takes precedence over wrinkling. On the other hand, hard-on-soft bilayer structures first grow wrinkles before transitioning into creases [Lee *et al.*, 2008; Wu *et al.*, 2013].

In addition to flat bilayer structures shown in Fig. 2, there have also been many studies on spherical bilayer structures, or core-shell structures. Spherical core-shell soft structures exhibit various morphologies, such as the buckyball, deformed polygons and labyrinth patterns [Fogle *et al.*, 2013; Komura *et al.*, 2005; Li *et al.*, 2011b; Yin *et al.*, 2008]. The study of such morphologies provides an insight on the growth of biological systems, such as tumors.



Fig. 2. Surface wrinkling of a bilayer structure. (a) Film attached to a substrate and (b) a core-shell structure.



Fig. 3. Surface creasing of a thick hydrogel layer.

#### 5.2.2. Surface instabilities of thick gel layers

When fixed to a rigid substrate, a thick gel layer with uniform [Ben Amar and Ciarletta, 2010; Hong *et al.*, 2009b; Trujillo *et al.*, 2008; Weiss *et al.*, 2013; Xiao *et al.*, 2012] and nonuniform crosslink densities [Guvendiren *et al.*, 2009; Wu *et al.*, 2013] experience surface instabilities, as illustrated in Fig. 3.

These surface instabilities are mainly in the form of creasing as stability analysis has shown that the onset of creases on the surface requires a considerably lower critical swelling ratio [Jin *et al.*, 2011; Jin *et al.*, 2014; Wong *et al.*, 2010]. Surface tension aids to suppress short wavelength instabilities [Kang and Huang, 2010a].

During swelling, wrinkles may appear during the transient process due to compressive stresses being induced. Over the time evolution, the stress distribution changes and thus causes the wrinkles to evolve from one wavelength to another. This phenomenon has been observed repeatedly in experiments [Guvendiren *et al.*, 2010; Peixinho and Mukhopadhyay, 2013; Tanaka *et al.*, 1992; Tanaka *et al.*, 1987]. Toh *et al.* [2015] modeled and investigated the evolution of wrinkle wavelengths using finite element and linear perturbation methods.

#### 5.2.3. Bulk buckling of thin gel films

Thin gels geometrically constrained at certain points buckle into periodic wrinkles during the swelling process. Figure 4 shows the wrinkling patterns of an annular gel and a gel strip. Experimentally and theoretically, it has been shown that the wavelength of the wrinkles depend on the geometry of the gel [Lee *et al.*, 2012; Li *et al.*, 2013a; Liu *et al.*, 2011; Mora and Boudaoud, 2006; Zhang *et al.*, 2014].



Fig. 4. Buckling into waves of different sizes of (a) gel annulus constrained on inner side and (b) gel strip constrained on 1 side.



Fig. 5. (a) Gel film initially with perfect circular geometry, (b) transits into elliptical holes with mutually perpendicular orientations, and (c) the elliptical holes close up to form mutually perpendicular slits.

#### 5.2.4. Bifurcation of gel films with square lattice of holes

In gel layers with periodic holes arranged in an array, the instability is not in the form of surface wrinkles. Instead, the holes bifurcate into alternating perpendicular slits (Fig. 5) [Bertoldi *et al.*, 2008; Mullin *et al.*, 2007; Okumura *et al.*, 2014].

Okumura *et al.* studied the effects of geometrical imperfections [Okumura *et al.*, 2014] and prestrain [Okumura *et al.*, 2015] on the resulting geometry. It was shown that while three buckling patterns were obtained, the diamond pattern was thermodynamically most favorable due to the lower stress level present.



Fig. 6. (a) The drying of a leaf, simulated by the dwelling of a hydrogel. (b) The growth of a leaf, simulated by the swelling of a hydrogel. (c) The formation of ripples in an apple, simulated by differential swelling between different layers of hydrogel [Liu *et al.*, 2010; 2013].

# 5.3. Explanation of natural phenomena

The use of soft materials to mimic natural phenomenon is getting more and more prevalent. Further, with the development of ABAQUS finite element subroutines, the instability of gels has been used to model and explain phenomena commonly observed in nature. Cao *et al.* [2012] likened the volumetric growth process to thermal stress to study wrinkle formation on skin. Amar and Goriely [2005] and Li *et al.* [2011a] modeled the growth process using an incremental deformation theory to study growth of soft tissues and mucosa, respectively.

On the other hand, Liu *et al.* [2010; 2013] made use of the volume change in hydrogels to model growth and the drying processes, which is a more realistic approach for studying the geometrical features of plants or phyllotaxis (as shown in Fig. 6). Similarly, Dervaux and Amar [2011] likened the growth of a tumor to the swelling of a hydrogel. This is achieved by modeling these items as hydrogels or soft materials, which are able to undergo volumetric expansion and shrinking, thus simulating the growth and drying processes and the associated buckling phenomena as energy minimizing mechanisms.

# 6. Applications for Soft Machines

In the emerging field of soft machines, hydrogels are the most viable and valuable candidate of soft materials that can provide extremely large deformation. Thus, with more research into smart hydrogels, the potential applications involving hydrogels has grown exponentially. Generally, individual standalone soft materials are not able to provide all of the requisite functions for soft machines; rather, soft machines are mostly hybrids of soft and hard materials. However, some hydrogels do possess the properties required for stretchable electronics applications, and can deform in response to stimuli other than mechanical forces.

# 6.1. Microfluidics

Hydrogels have been shown to be able to function well as flow control devices [Baldi et al., 2002; Harmon et al., 2003; Johnson et al., 2004; Lai et al., 2007; Moore et al., 2000; Satarkar et al., 2009; Sugiura et al., 2007; Wang et al., 2005] in micro systems. Various configurations and designs have been proposed, but a common technique is making use of the contact stress between a swollen gel and a rigid structure to restrict flow movement across a channel. Figure 7 shows two types of mechanisms in which hydrogels are utilized in microfluidic devices.

Through understanding of gel mechanics, it is possible to better design such applications. Zhang *et al.* [2012] and He *et al.* [2012] modeled pH-sensitive hydrogel valves, studying the fluid-structure interaction and contact stresses within these valves.

# 6.2. Self-assembled structures

With localized application of external stimuli, it is possible to program a gel to swell and fold in a certain way, thus forming self-folding smart structures. Ryu *et al.* [2012] introduced a self-folding box using localized light absorption and Na *et al.* [2015] demonstrated a reversible self-folding crane using trilayer films.



Fig. 7. The examples of hydrogel actuated microfluidic valves.

The swelling induced buckling of gels can be harnessed for a novel way to create gear systems [Yin *et al.*, 2009; Zhang *et al.*, 2010]. More recently, Hu *et al.* discovered a negative Poisson ratio phenomenon in soft materials, which holds promising applications for novel structures [Hu *et al.*, 2013b; Hu *et al.*, 2014].

# 6.3. Sensors and actuators

When bound to a substrate, a gel structure undergoes differential swelling and bends as a result. This effect is useful for actuation [Bassik *et al.*, 2010; Hoffmann *et al.*, 1999; Hong *et al.*, 2008a; Kim *et al.*, 2011; Wang *et al.*, 2013]. As compared to traditional actuators made from more conventional engineering materials, gel actuators allow for a larger extent of actuation due to the larger deformations.

Other than machines, hydrogel actuators also hold a promising future as artificial muscles due to their bio-compatibility [Bassil *et al.*, 2008; Bassil *et al.*, 2011; Mirfakhrai *et al.*, 2007].

The large deformation, coupled with environmentally responsive properties of certain types of gels, also make for very good sensors [Bashir *et al.*, 2002; Richter *et al.*, 2008; Sheppard Jr. *et al.*, 1995; van der Linden *et al.*, 2003; Zhang *et al.*, 2004].

# 6.4. Tunable lens

With the prevalence and advancement of mobile phone cameras, the trend of taking photos is slowly evolving toward phones replacing cameras. Due to their cumbersome implementation, removable hard lens are losing popularity. This compromises on the quality of photographs being taken. However, soft lens are able to solve this problem, as the degree of curvature and thickness of a gel lens can easily be altered using external stimuli, such as temperature [Zeng *et al.*, 2010] and pH-value [Zalachas *et al.*, 2012].

Due to large deformation, creases may form in the microlens. Zalachas *et al.* [2013] modeled the creasing instability that arises in such configurations.

# 6.5. Drug delivery

Making use of characteristic properties of different parts of a body, hydrogels can be customized to target specific areas for drug release [Chu, 2003; Elvira *et al.*, 2004; Giani *et al.*, 2012; Gupta *et al.*, 2002; Jeong *et al.*, 2006; Liu *et al.*, 2006; Qiu and Park, 2001; Satarkar and Hilt, 2008; Siepmann *et al.*, 1999; Uva *et al.*, 2014; Zhao *et al.*, 2010]. In the drug delivery process, it is imperative that the deformation kinetics be fully understood.

# 6.6. Hydrogel ionic conductors

Hydrogel ionic conductors make it possible to develop new stretchable electronic devices. The Harvard group has developed a class of new devices enabled by ionic conductors. In these devices, the ionic conductors are highly stretchable, and fully

transparent to light of all colors [Keplinger *et al.*, 2013]. In their study, they reported that the transparent actuator can generate large strains and also act as a transparent loudspeaker. They highlighted that in the device, the electromechanical transduction can work without any electrochemical reaction. This achievement based on ionic conductors is advantageous over the existing stretchable, transparent electronic conductors. Their findings provide possibilities of an alternative stretchable ionic conductor based on a deformable hydrogel device.

# 7. Potential Research Directions of the Mechanics of Hydrogels

Much work remains to be done in the area of mechanics of hydrogels. First, in the area of modeling and simulation, more efforts can be devoted to develop more robust simulation tools to predict material response when exposed to a diverse array of external stimuli.

While much has been achieved in the modeling and simulation of hydrogels, there are seemingly less experimental works focusing on the characterization of mechanical behaviors of hydrogels.

The large deformation present in soft materials makes it an attractive material for replacement of traditional engineering materials. As seen in Sec. 6, there is a trend of machines moving away from hard engineering materials and taking the form of soft machines.

In materials engineering, stronger and better materials systems or composites are obtained when we combine materials of different properties. This notion of developing superior materials can also be extended to soft materials, such as gel and shape memory polymer (SMP) composite materials [He *et al.*, 2015]. Due to the excellent compatibility of hydrogel-based composites, and their enhanced deformation behavior and mechanical properties, more advanced combined materials are emerging especially in soft machine design.

As many issues related to the mechanics of hydrogel behaviors remain open, we list below some outlines for plausible future directions in the research of the mechanics of hydrogels.

- (1) In the modeling and simulation of deformation behavior of hydrogel, more robust kinetic models and more robust subroutines which can simulate the diffusion mechanism of solvent molecules and ions over a long timeframe, as well as short timeframe effects, such as viscoelastic effects, needs to be further developed [Toh, 2015].
- (2) It will be useful to develop a unified constitutive model for hydrogels deformation and to study the effects of various stimuli on the gel. This unified model can consider different stimuli, such as mechanical, temperature, photonics, chemical, electrical stimuli, as well as the coupling of different stimuli.
- (3) Current study of phase transition phenomenon of hydrogels is limited to the equilibrium swelling state with homogenous dispersion of stimuli-induced

chemical reactions within the gel, such as photo-chemical reactions. However, the distribution of such chemical reactions in gels is never homogeneous. Therefore, it will be imperative to develop the models which can incorporate this inhomogeneity.

- (4) Soft machines are mostly hybrids of soft and hard materials, for example devices based on the fabrication of islands consisting of stiff materials on polymer (hydrogel) substrates. In this design, many issues concerning the mechanics of hydrogels will emerge, such as the deformation compatibility of soft materials and hard materials and their interaction, as well as the contact and adhesion issues between the different materials. Therefore, the various deformation modes of wrinkling, twisting, buckling and necking of hydogels or combined hydrogels bilayers should be investigated further.
- (5) Many parts of the human body are hydrogel-like in nature, and can be assumed to be mixtures of macromolecules and water. The meniscus is one such example, which can be repaired by using special hydrogels. Thus, the mechanics of hydrogels will become an important issue in biomechanics as hydrogels find more applications in the human body.

# 8. Conclusion

This paper has been written with the aim to consolidate ideas related to the mechanics of hydrogels: the theories developed, the phenomena associated and the applications. Both multiphase and monophase theories have been explored, with attention being cast more on the monophase theories for its unifying approach and relative ease of use in developing simulation models.

With better understanding of the mechanics of hydrogels, it is possible to develop novel applications which may possess superior performance over current conventional engineering materials. Furthermore, the similarity between hydrogel swelling and morphologies in nature allows us to make use of gel mechanics to gain a deeper insight into our natural world.

# Acknowledgments

The authors are grateful for the support from the National Natural Science Foundation of China through Grant Nos. 11372236, 11572236 and 11321062. The authors also would like to express their appreciation to Nanyang Technological University (NTU).

# References

Afroze, F., Nies, E. and Berghmans, H. [2000] "Phase transitions in the system poly(Nisopropylacrylamide)/water and swelling behaviour of the corresponding networks," *Journal of Molecular Structure* 554, 55–68.

- Aguilar, M. R., Elvira, C., Gallardo, A. and Roman, J. S. [2007] "Smart polymers and their applications as biomaterials," *Topics in Tissue Engineering*.
- Amiya, T., Hirokawa, Y., Hirose, Y., Li, Y. and Tanaka, T. [1987] "Reentrant phase transition of N-isopropylacrylamide gels in mixed solvents," *The Journal of Chemical Physics* 86, 2375–2379.
- An, Y., Solis, F. J. and Jiang, H. [2010] "A thermodynamic model of physical gels," Journal of the Mechanics and Physics of Solids 58, 2083–2099.
- Anand, L. [1996] "A constitutive model for compressible elastomeric solids," Computational Mechanics 18, 339–355.
- Arruda, E. M. and Boyce, M. C. [1993] "A three-dimensional constitutive model for the large stretch behavior of rubber elastic materials," *Journal of the Mechanics and Physics of Solids* 41, 389–412.
- Baek, S. and Pence, T. J. [2011] "Inhomogeneous deformation of elastomer gels in equilibrium under saturated and unsaturated conditions," *Journal of the Mechanics and Physics of Solids* 59, 561–582.
- Baek, S. and Srinivasa, A. R. [2004] "Diffusion of a fluid through an elastic solid undergoing large deformation," *International Journal of Non-Linear Mechanics* 39, 201–218.
- Baldi, A., Yuandong, G., Loftness, P. E., Siegel, R. A. and Ziaie, B. [2002] "A hydrogelactuated smart microvalve with a porous diffusion barrier back-plate for active flow control," *The Fifteenth IEEE International Conference on Micro Electro Mechanical* Systems, 2002, pp. 105–108.
- Barros, W., de Azevedo, E. N. and Engelsberg, M. [2012] "Surface pattern formation in a swelling gel," Soft Matter 8, 8511–8516.
- Bashir, R., Hilt, J. Z., Elibol, O., Gupta, A. and Peppas, N. A. [2002] "Micromechanical cantilever as an ultrasensitive pH microsensor," *Applied Physics Letters* 81, 3091–3091.
- Bassik, N., Abebe, B. T., Laflin, K. E. and Gracias, D. H. [2010] "Photolithographically patterned smart hydrogel based bilayer actuators," *Polymer* 51, 6093–6098.
- Bassil, M., Davenas, J. and El Tahchi, M. [2008] "Electrochemical properties and actuation mechanisms of polyacrylamide hydrogel for artificial muscle application," *Sensors and Actuators B: Chemical* 134, 496–501.
- Bassil, M., Ibrahim, M. and El Tahchi, M. [2011] "Artificial muscular microfibers: Hydrogel with high speed tunable electroactivity," Soft Matter 7, 4833–4838.
- Ben Amar, M. and Ciarletta, P. [2010] "Swelling instability of surface-attached gels as a model of soft tissue growth under geometric constraints," *Journal of the Mechanics* and Physics of Solids 58, 935–954.
- Ben Amar, M. and Goriely, A. [2005] "Growth and instability in elastic tissues," Journal of the Mechanics and Physics of Solids 53, 2284–2319.
- Bertoldi, K., Boyce, M. C., Deschanel, S., Prange, S. M. and Mullin, T. [2008] "Mechanics of deformation-triggered pattern transformations and superelastic behavior in periodic elastomeric structures," *Journal of the Mechanics and Physics of Solids* 56, 2642–2668.
- Birgersson, E., Li, H. and Wu, S. [2008] "Transient analysis of temperature-sensitive neutral hydrogels," *Journal of the Mechanics and Physics of Solids* 56, 444–466.
- Bouklas, N. and Huang, R. [2012] "Swelling kinetics of polymer gels: Comparison of linear and nonlinear theories," Soft Matter 8, 8194–8203.
- Boyce, M. C. and Arruda, E. M. [2001] "Swelling and mechanical stretching of elastomeric materials," *Mathematics and Mechanics of Solids* 6, 641–659.
- Boyce, M. C. and Arruda, E. M. [2000] "Constitutive models of rubber elasticity: A review," *Rubber Chemistry and Technology* **73**, 504–523.
- Brannon-Peppas, L. and Peppas, N. A. [1991] "Equilibrium swelling behavior of pHsensitive hydrogels," *Chemical Engineering Science* 46, 715–722.

- Cai, S., Breid, D., Crosby, A. J., Suo, Z. and Hutchinson, J. W. [2011] "Periodic patterns and energy states of buckled films on compliant substrates," *Journal of the Mechanics* and Physics of Solids 59, 1094–1114.
- Cai, S. and Suo, Z. [2011] "Mechanics and chemical thermodynamics of phase transition in temperature-sensitive hydrogels," *Journal of the Mechanics and Physics of Solids* 59, 2259–2278.
- Cai, S. and Suo, Z. [2012] "Equations of state for ideal elastomeric gels," *Europhysics Letters* 97, 34009.
- Calderer, M. C., Chabaud, B., Lyu, S. and Zhang, H. [2008] "Modeling approaches to the dynamics of hydrogel swelling," *Journal of Computational and Theoretical Nanoscience* 7, 766–779.
- Calvert, P. [2009] "Hydrogels for soft machines," Advanced Materials 21, 743-756.
- Cao, Y., Jiang, Y., Li, B. and Feng, X. [2012] "Biomechanical modeling of surface wrinkling of soft tissues with growth-dependent mechanical properties," Acta Mechanica Solida Sinica 25, 483–492.
- Caykara, T., Kiper, S. and Demirel, G. [2006] "Network parameters and volume phase transition behavior of poly(N-isopropylacrylamide) hydrogels," *Journal of Applied Polymer Science* 101, 1756–1762.
- Chen, C.-M. and Yang, S. [2012] "Wrinkling instabilities in polymer films and their applications," *Polymer International* 61, 1041–1047.
- Chen, J., Li, H. and Lam, K. Y. [2005] "Transient simulation for kinetic responsive behaviors of electric-sensitive hydrogels subject to applied electric field," *Materials Science* and Engineering: C 25, 710–712.
- Chen, X. and Yin, J. [2010] "Buckling patterns of thin films on curved compliant substrates with applications to morphogenesis and three-dimensional micro-fabrication," *Soft Matter* 6, 5667–5680.
- Chester, S. A. [2012] "A constitutive model for coupled fluid permeation and large viscoelastic deformation in polymeric gels," Soft Matter 8, 8223–8233.
- Chester, S. A. and Anand, L. [2010] "A coupled theory of fluid permeation and large deformations for elastomeric materials," *Journal of the Mechanics and Physics of Solids* 58, 1879–1906.
- Chester, S. A. and Anand, L. [2011] "A thermo-mechanically coupled theory for fluid permeation in elastomeric materials: Application to thermally responsive gels," *Journal* of the Mechanics and Physics of Solids 59, 1978–2006.
- Chester, S. A., Di Leo, C. V. and Anand, L. [2014] "A finite element implementation of a coupled diffusion-deformation theory for elastomeric gels," *International Journal of Solids and Structures*
- Chu, C. [2003] Biodegradable Hydrogels as Drug Controlled Release Vehicles, Tissue Engineering and Novel Delivery Systems (CRC Press, Cornell University, Ithaca, New York).
- Corkhill, P. H., Hamilton, C. J. and Tighe, B. J. [1989] "Synthetic hydrogels VI. Hydrogel composites as wound dressings and implant materials," *Biomaterials* 10, 3–10.
- De, S. K. and Aluru, N. R. [2004] "A chemo-electro-mechanical mathematical model for simulation of pH sensitive hydrogels," *Mechanics of Materials* 36, 395–410.
- De, S. K., Aluru, N. R., Johnson, B., Crone, W. C., Beebe, D. J. and Moore, J. [2002] "Equilibrium swelling and kinetics of pH-responsive hydrogels: Models, experiments, and simulations," *Journal of Microelectromechanical Systems* 11, 544– 555.
- Dervaux, J. and Ben Amar, M. [2011] "Buckling condensation in constrained growth," Journal of the Mechanics and Physics of Solids 59, 538–560.

- Ding, Z., Liu, Z. S., Hu, J., Swaddiwudhipong, S. and Yang, Z. [2013] "Inhomogeneous large deformation study of temperature-sensitive hydrogel," *International Journal of Solids and Structures* 50, 2610–2619.
- Ding, Z. W., Toh, W., Liu, Z. S. and Ng, T. Y. [Under review] "A Coupled Thermo-Mechanical Model for the Transient Analysis of Temperature-Sensitive Hydrogels.
- Doi, M. [2009] "Gel dynamics," Journal of the Physical Society of Japan 78, 052001– 052001.
- Doi, M. and Onuki, A. [1992] "Dynamic coupling between stress and composition in polymer solution and blends," *Journal of Physics II (Paris)* 2, 1631–1656.
- Dolbow, J., Fried, E. and Ji, H. [2004] "Chemically induced swelling of hydrogels," Journal of the Mechanics and Physics of Solids 52, 51–84.
- Dolbow, J., Fried, E. and Ji, H. [2005] "A numerical strategy for investigating the kinetic response of stimulus-responsive hydrogels," *Computer Methods in Applied Mechanics* and Engineering 194, 4447–4480.
- Douezan, S., Wyart, M., Brochard-Wyart, F. and Cuvelier, D. [2011] "Curling instability induced by swelling," Soft Matter 7, 1506–1511.
- Drozdov, A. D. [2014a] "Swelling of thermo-responsive hydrogels," European Physics Journal E 37, 1–13.
- Drozdov, A. D. [2014b] "Volume phase transition in thermo-responsive hydrogels: Constitutive modeling and structure-property relations," Acta Mechanica 226, 1283–1303.
- Duan, Z., Zhang, J., An, Y. and Jiang, H. [2013] "Simulation of the transient behavior of gels based on an analogy between diffusion and heat transfer," *Journal of Applied Mechanics* 80, 041017–041017.
- Duda, F. P., Souza, A. C. and Fried, E. [2010] "A theory for species migration in a finitely strained solid with application to polymer network swelling," *Journal of the Mechanics* and Physics of Solids 58, 515–529.
- DuPont, Jr., S. J., Cates, R. S., Stroot, P. G. and Toomey, R. [2010] "Swelling-induced instabilities in microscale, surface-confined poly(N-isopropylacryamide) hydrogels," *Soft Matter* 6, 3876–3882.
- Durning, C. J. and Morman, K. N. J. [1993] "Nonlinear swelling of polymer gels," The Journal of Chemical Physics 98, 4275–4293.
- Elvira, C., Abraham, G., Gallardo, A., Roman, J. S. and Román, J. S. [2004] Smart Biodegradable Hydrogels with Applications in Drug Delivery and Tissue Engineering (CRC Press, Boca Raton, FL).
- Erman, B. and Flory, P. J. [1986] "Critical phenomena and transitions in swollen polymer networks and in linear macromolecules," *Macromolecules* 19, 2342–2353.
- Flory, P. J. [1942] "Thermodynamics of high polymer solutions," The Journal of Chemical Physics 10, 51–61.
- Flory, P. J. [1953] Principles of Polymer Chemistry (Cornell University Press, Ithaca).
- Flory, P. J. and Rehner, J. [1943] "Statistical mechanics of cross-linked polymer networks I: Rubberlike elasticity," *The Journal of Chemical Physics* 11, 512–520.
- Fogle, C., Rowat, A. C., Levine, A. J. and Rudnick, J. [2013] "Shape transitions in soft spheres regulated by elasticity," *Physical Review E* 88, 052404.
- Giani, G., Fedi, S. and Barbucci, R. [2012] "Hybrid magnetic hydrogel: A potential system for controlled drug delivery by means of alternating magnetic fields," *Polymers* 4, 1157–1169.
- Gong, J. P. [2010] "Why are double network hydrogels so tough?" Soft Matter 6, 2583–2590.
- Gong, J. P., Katsuyama, Y., Kurokawa, T. and Osada, Y. [2003] "Double-network hydrogels with extremely high mechanical strength," Advanced Materials 15, 1155–1158.

- Gupta, P., Vermani, K. and Garg, S. [2002] "Hydrogels: From controlled release to pHresponsive drug delivery," Drug Discovery Today 7, 569–579.
- Guvendiren, M., Burdick, J. A. and Yang, S. [2010] "Kinetic study of swelling-induced surface pattern formation and ordering in hydrogel films with depth-wise crosslinking gradient," Soft Matter 6, 2044–2049.
- Guvendiren, M., Yang, S. and Burdick, J. A. [2009] "Swelling-induced surface patterns in hydrogels with gradient crosslinking density," *Advanced Functional Materials* 19, 3038–3045.
- Han, Y. I., Hong, W. E. I. and Faidley, L. [2011] "Coupled magnetic field and viscoelasticity of ferrogel," *International Journal of Applied Mechanics* 3, 259–278.
- Harmon, M. E., Tang, M. and Frank, C. W. [2003] "A microfluidic actuator based on thermoresponsive hydrogels," *Polymer* 44, 4547–4556.
- Hassan, M. M. and Durning, C. J. [1999] "Effects of polymer molecular weight and temperature on case II transport," *Journal of Polymer Science Part B: Polymer Physics* 37, 3159–3171.
- He, T., Li, M. and Zhou, J. [2012] "Modeling deformation and contacts of pH sensitive hydrogels for microfluidic flow control," *Soft Matter* 8, 3083–3089.
- He, Y., Guo, S., Liu, Z. and Liew, K. M. [2015] "Pattern transformation of thermoresponsive shape memory polymer periodic cellular structures," *International Journal* of Solids and Structures 71, 194–205.
- Helfferich, F. and Plesset, M. S. [1958] "Ion exchange kinetics: A nonlinear diffusion problem," The Journal of Chemical Physics 28, 418–424.
- Hino, T. and Prausnitz, J. M. [1998] "Molecular thermodynamics for volume-change transitions in temperature-sensitive polymer gels," *Polymer* 39, 3279–3283.
- Hirokawa, Y. and Tanaka, T. [1984] "Volume phase transition in a nonionic gel," The Journal of Chemical Physics 81, 6379–6380.
- Hirose, Y., Amiya, T., Hirokawa, Y. and Tanaka, T. [1987] "Phase transition of submicron gel beads," *Macromolecules* 20, 1342–1344.
- Hirotsu, S., Hirokawa, Y. and Tanaka, T. [1987] "Volume: Phase transitions of ionized N-isopropylacrylamide gels," *The Journal of Chemical Physics* 87, 1392–1395.
- Hoffmann, J., Plötner, M., Kuckling, D. and Fischer, W.-J. [1999] "Photopatterning of thermally sensitive hydrogels useful for microactuators," *Sensors and Actuators A: Physical* 77, 139–144.
- Hong, W., Liu, Z. S. and Suo, Z. [2009a] "Inhomogeneous swelling of a gel in equilibrium with a solvent and mechanical load," *International Journal of Solids and Structures* 46, 3282–3289.
- Hong, W., Zhao, X. and Suo, Z. [2008a] "Drying-induced bifurcation in a hydrogel-actuated nanostructure," *Journal of Applied Physics* 104, 084905–084905.
- Hong, W., Zhao, X. and Suo, Z. [2009b] "Formation of creases on the surfaces of elastomers and gels," *Applied Physics Letters* 95, 111901.
- Hong, W., Zhao, X. and Suo, Z. [2010] "Large deformation and electrochemistry of polyelectrolyte gels," *Journal of the Mechanics and Physics of Solids* 58, 558–577.
- Hong, W., Zhao, X., Zhou, J. and Suo, Z. [2008b] "A theory of coupled diffusion and large deformation in polymeric gels," *Journal of the Mechanics and Physics of Solids* 56, 1779–1793.
- Hu, J., Ding, Z. and Liu, Z. S. [2013a] "Large deformation study of temperature-sensitive hydrogel," 13th International Conference on Fracture, Beijing, China.
- Hu, J., He, Y., Lei, J. and Liu, Z. [2013b] "Novel mechanical behavior of periodic structure with the pattern transformation," *Theoretical and Applied Mechanics Letters* 3, 054007.

- Hu, J. Y., He, Y. H., Lei, J. C., Liu, Z. S. and Swaddiwudhipong, S. [2014] "Mechanical behavior of composite gel periodic structures with the pattern transformation," *Structural Engineering and Mechanics* 50, 605–616.
- Hu, Z., Chen, Y., Wang, C., Zheng, Y. and Li, Y. [1998] "Polymer gels with engineered environmentally responsive surface patterns," *Nature* 393, 149–152.
- Huang, R. and Suo, Z. [2002a] "Instability of a compressed elastic film on a viscous layer," International Journal of Solids and Structures 39, 1791–1802.
- Huang, R. and Suo, Z. [2002b] "Wrinkling of a compressed elastic film on a viscous layer," *Journal of Applied Physics* 91, 1135–1142.
- Huang, Z., Hong, W. and Suo, Z. [2004] "Evolution of wrinkles in hard films on soft substrates," *Physical Review E* 70, 30601.
- Huggins, M. L. [1941] "Solutions of long chain compounds," The Journal of Chemical Physics 9, 440.
- Huggins, M. L. [1964] "A revised theory of high polymer solutions," Journal of the American Chemical Society 86, 3535–3540.
- Jaramillo-Botero, A., Blanco, M., Li, Y., McGuinness, G. and Goddard, W. A. [2010] "First-principles based approaches to nano-mechanical and biomimetic characterization of polymer-based hydrogel networks for cartilage scaffold-supported therapies," *Journal of Computational and Theoretical Nanoscience* 7, 1238–1256.
- Jeong, B. and Gutowska, A. [2002] "Lessons from nature: stimuli-responsive polymers and their biomedical applications," *Trends Biotechnology* 20, 305–311.
- Jeong, S. H., Huh, K. M. and Park, K. [2006] Hydrogel Drug Delivery Systems (CRC Press, Boca Raton, FL).
- Ji, H., Mourad, H., Fried, E. and Dolbow, J. [2006] "Kinetics of thermally induced swelling of hydrogels," *International Journal of Solids and Structures* 43, 1878–1907.
- Ji, S. and Ding, J. [2002] "The wetting process of a dry polymeric hydrogel," Polymer Journal 34, 267–270.
- Jia, F. and Ben Amar, M. [2013] "Theoretical analysis of growth or swelling wrinkles on constrained soft slabs," Soft Matter 9, 8216–8226.
- Jin, L., Cai, S. and Suo, Z. [2011] "Creases in soft tissues generated by growth," *Europhysics Letters* 95, 64002.
- Jin, L., Chen, D., Hayward, R. C. and Suo, Z. [2014] "Creases on the interface between two soft materials," Soft Matter 10, 303–311.
- Johnson, B. D., Beebe, D. J. and Crone, W. C. [2004] "Effects of swelling on the mechanical properties of a pH-sensitive hydrogel for use in microfluidic devices," *Materials Science* and Engineering: C 24, 575–581.
- Juodkazis, S., Mukai, N., Wakaki, R., Yamaguchi, A., Matsuo, S. and Misawa, H. [2000] "Reversible phase transitions in polymer gels induced by radiation forces," *Nature* 408, 178–181.
- Kang, M. K. and Huang, R. [2010a] "Effect of surface tension on swell-induced surface instability of substrate-confined hydrogel layers," Soft Matter 6, 5736–5742.
- Kang, M. K. and Huang, R. [2010b] "Swell-induced surface instability of confined hydrogel layers on substrates," *Journal of the Mechanics and Physics of Solids* 58, 1582–1598.
- Kang, M. K. and Huang, R. [2010c] "A variational approach and finite element implementation for swelling of polymeric hydrogels under geometric constraints," *Journal* of Applied Mechanics 77, 61004–61012.
- Kenkare, N. R., Hall, C. K. and Khan, S. A. [2000] "Theory and simulation of the swelling of polymer gels," *The Journal of Chemical Physics* 113, 404–418.
- Keplinger, C., Sun, J.-Y., Foo, C. C., Rothemund, P., Whitesides, G. M. and Suo, Z. [2013] "Stretchable, transparent, ionic conductors," *Science* 341, 984–987.

- Kim, J., Yoon, J. and Hayward, R. C. [2010] "Dynamic display of biomolecular patterns through an elastic creasing instability of stimuli-responsive hydrogels," *Nature Materials* 9, 159–164.
- Kim, P., Zarzar, L. D., He, X., Grinthal, A. and Aizenberg, J. [2011] "Hydrogel-actuated integrated responsive systems (HAIRS): Moving towards adaptive materials," *Current Opinion in Solid State and Materials Science* 15, 236–245.
- Klein, Y., Efrati, E. and Sharon, E. [2007] "Shaping of elastic sheets by prescription of non-euclidean metrics," *Science* **315**, 1116–1120.
- Kojima, H., Tanaka, F., Scherzinger, C. and Richtering, W. [2013] "Temperature dependent phase behavior of PNIPAM microgels in mixed water/methanol solvents," *Journal* of Polymer Science Part B: Polymer Physics 51, 1100–1111.
- Komura, S., Tamura, K. and Kato, T. [2005] "Buckling of spherical shells adhering onto a rigid substrate," *European Physics Journal E: Soft Matter* 18, 343–358.
- Kondo, O., Suzuki, A. and Shimazaki, T. [1993] "Volume phase transition in polymer gels induced by uniaxial stress," *Journal of Intelligent Material Systems and Structures* 4, 279–282.
- Kuroyanagi, Y. [1999] "Advances in wound dressings and cultured skin substitutes," Journal of Artificial Organs 2, 97–116.
- Kwon, G. H., Choi, Y. Y., Park, J. Y., Woo, D. H., Lee, K. B., Kim, J. H. and Lee, S.-H. [2010] "Electrically-driven hydrogel actuators in microfluidic channels: Fabrication, characterization, and biological application," *Lab on a Chip* 10, 1604–1610.
- Lai, F. and Li, H. [2010] "Transient modeling for kinetic swelling/deswelling of the ionicstrength-sensitive hydrogel," *European Physics Journal E* 31, 269–274.
- Lai, F. and Li, H. [2011] "Transient modeling of the reversible response of the hydrogel to the change in the ionic strength of solutions," *Mechanics of Materials* **43**, 287–298.
- Lai, J. J., Hoffman, J. M., Ebara, M., Hoffman, A. S., Estournès, C., Wattiaux, A. and Stayton, P. S. [2007] "Dual magnetic-/temperature-responsive nanoparticles for microfluidic separations and assays," *Langmuir* 23, 7385–7391.
- Lam, K. Y., Li, H., Ng, T. Y. and Luo, R. [2006] "Modeling and simulation of the deformation of multi-state hydrogels subjected to electrical stimuli," *Engineering Analysis* with Boundary Elements 30, 1011–1017.
- Lee, D., Triantafyllidis, N., Barber, J. R. and Thouless, M. D. [2008] "Surface instability of an elastic half space with material properties varying with depth," *Journal of the Mechanics and Physics of Solids* 56, 858–868.
- Lee, E., Zhang, M., Cho, Y., Cui, Y., Van der Spiegel, J., Engheta, N. and Yang, S. [2014] "Tilted pillars on wrinkled elastomers as a reversibly tunable optical window," *Advanced Materials* 26, 4127–4133.
- Lee, H., Zhang, J., Jiang, H. and Fang, N. X. [2012] "Prescribed pattern transformation in swelling gel tubes by elastic instability," *Physical Review Letters* 108, 214304–214304.
- Li, B., Cao, Y.-P., Feng, X.-Q. and Gao, H. [2011a] "Surface wrinkling of mucosa induced by volumetric growth: Theory, simulation and experiment," *Journal of the Mechanics* and Physics of Solids 59, 758–774.
- Li, B., Jia, F., Cao, Y.-P., Feng, X.-Q. and Gao, H. [2011b] "Surface wrinkling patterns on a core-shell soft sphere," *Physical Review Letters* **106**, 234301.
- Li, D. M., Zhang, Z. and Liew, K. M. [2014] "A numerical framework for two-dimensional large deformation of inhomogeneous swelling of gels using the improved complex variable element-free Galerkin method," *Computer Methods in Applied Mechanics and Engineering* 274, 84–102.
- Li, H. [2009] "Kinetics of smart hydrogels responding to electric field: A transient deformation analysis," *International Journal of Solids and Structures* 46, 1326–1333.

- Li, H., Chen, J. and Lam, K. Y. [2004] "Multiphysical modeling and meshless simulation of electric-sensitive hydrogels," *Journal of Polymer Science Part B: Polymer Physics* 42, 1514–1531.
- Li, H., Chen, J. and Lam, K. Y. [2006] "A transient simulation to predict the kinetic behavior of hydrogels responsive to electric stimulus," *Biomacromolecules* 7, 1951– 1959.
- Li, H., Chen, J. and Lam, K. Y. [2007a] "Transient simulation of kinetics of electricsensitive hydrogels," *Biosensors and Bioelectronics* 22, 1633–1641.
- Li, H. and Lai, F. [2011] "Transient analysis of the effect of the initial fixed charge density on the kinetic characteristics of the ionic-strength-sensitive hydrogel by a multi-effectcoupling model," Analytical and Bioanalytical Chemistry 399, 1233–1243.
- Li, H., Luo, R., Birgersson, E. and Lam, K. Y. [2007b] "Modeling of multiphase smart hydrogels responding to pH and electric voltage coupled stimuli," *Journal of Applied Physics* 101, 114905.
- Li, H., Luo, R. and Lam, K. Y. [2007c] "Modeling and simulation of deformation of hydrogels responding to electric stimulus," *Journal of Biomechanics* 40, 1091–1098.
- Li, H. and Mulay, S. S. [2011] "2D simulation of the deformation of pH-sensitive hydrogel by novel strong-form meshless random differential quadrature method," *Computational Mechanics* 48, 729–753.
- Li, H., Ng, T. Y., Yew, Y. K. and Lam, K. Y. [2005a] "Modeling and simulation of the swelling behavior of pH-stimulus-responsive hydrogels," *Biomacromolecules* 6, 109– 120.
- Li, H., Wang, X., Yan, G., Lam, K. Y., Cheng, S., Zou, T. and Zhuo, R. [2005b] "A novel multiphysic model for simulation of swelling equilibrium of ionized thermal-stimulus responsive hydrogels," *Chemical Physics* **309**, 201–208.
- Li, H. and Yew, Y. K. [2009] "Simulation of soft smart hydrogels responsive to pH stimulus: Ionic strength effect and case studies," *Materials Science and Engineering: C* 29, 2261– 2269.
- Li, H., Yew, Y. K., Lam, K. Y. and Ng, T. Y. [2004] "Numerical simulation of pH-stimuli responsive hydrogel in buffer solutions," *Colloids and Surfaces A: Physicochemical and Engineering Aspects* 249, 149–154.
- Li, H., Yew, Y. K., Ng, T. Y. and Lam, K. Y. [2005c] "Meshless steady-state analysis of chemo-electro-mechanical coupling behavior of pH-sensitive hydrogel in buffered solution," *Journal of Electroanalytical Chemistry* 580, 161–172.
- Li, J., Hu, Y., Vlassak, J. J. and Suo, Z. [2012] "Experimental determination of equations of state for ideal elastomeric gels," *Soft Matter*.
- Li, K., Ding, K. and Cai, S. [2013a] "Diffusion-induced wrinkling instability in a circular poroelastic plate," Applied Physics Letters 102, 1–4.
- Li, M., Jin, C. and Zhou, J. [2013b] "Finite element implementation of poroelasticity theory for swelling dynamics of hydrogels," *Theoretical and Applied Mechanics Letters* 3.
- Li, M. and Kong, J. [2007] Smart Hydrogels, Smart Polymers (CRC Press, Boca Raton, FL), pp. 247–268.
- Li, Y. and Tanaka, T. [1990] "Kinetics of swelling and shrinking of gels," The Journal of Chemical Physics 92, 1365–1371.
- Liang, H. and Mahadevan, L. [2011] "Growth, geometry, and mechanics of a blooming lily," *Proceedings of the National Academy of Sciences* 108, 5516–5521.
- Liu, T.-Y., Hu, S.-H., Liu, D.-M. and Chen, S.-Y. [2006] "Magnetic-sensitive behavior of intelligent ferrogels for controlled release of drug," *Langmuir* 22, 5974–5978.

- Liu, Y., Zhang, H. and Zheng, Y. [2015] "A multiplicative finite element algorithm for the inhomogeneous swelling of polymeric gels," *Computer Methods in Applied Mechanics* and Engineering 283, 517–550.
- Liu, Z. S., Hong, W., Suo, Z., Swaddiwudhipong, S. and Zhang, Y. [2010] "Modeling and simulation of buckling of polymeric membrane thin film gel," *Computational Materials Science* 49, S60–S64.
- Liu, Z. S., Swaddiwudhipong, S., Cui, F. S., Hong, W., Suo, Z. and Zhang, Y. W. [2011] "Analytical solutions of polymeric gel structures under buckling and wrinkle," *International Journal of Applied Mechanics* 3, 235–257.
- Liu, Z. S., Swaddiwudhipong, S. and Hong, W. [2013] "Pattern formation in plants via instability theory of hydrogels," *Soft Matter* 9, 577–587.
- Lopez-Pamies, O. [2010] "A new -based hyperelastic model for rubber elastic materials," Comptes Rendus Mécanique 338, 3–11.
- Lucantonio, A., Nardinocchi, P. and Teresi, L. [2013] "Transient analysis of swellinginduced large deformations in polymer gels," *Journal of the Mechanics and Physics* of Solids 61, 205–218.
- Luo, R., Li, H. and Lam, K. Y. [2007] "Coupled chemo-electro-mechanical simulation for smart hydrogels that are responsive to an external electric field," *Smart Materials and Structures* 16, 1185.
- Luo, R., Li, H. and Lam, K. Y. [2008] "Modeling and analysis of pH-electric-stimuliresponsive hydrogels," *Journal of Biomaterials Science, Polymer Edition* 19, 1597– 1610.
- Mamada, A., Tanaka, T., Kungwatchakun, D. and Irie, M. [1990] "Photoinduced phase transition of gels," *Macromolecules* 23, 1517–1519.
- Marckmann, G. and Verron, E. [2006] "Comparison of hyperelastic models for rubber-like materials," *Rubber Chemistry and Technology* 79, 835–858.
- Marcombe, R., Cai, S., Hong, W., Zhao, X., Lapusta, Y. and Suo, Z. [2010] "A theory of constrained swelling of a pH-sensitive hydrogel," Soft Matter 6, 784–793.
- Matsuo, E. S. and Tanaka, T. [1992] "Patterns in shrinking gels," Nature 358, 482–485.
- Meng, H. and Hu, J. [2010] "A brief review of stimulus-active polymers responsive to thermal, light, magnetic, electric, and water/solvent stimuli," *Journal of Intelligent Material Systems and Structures* 21, 859–885.
- Mirfakhrai, T., Madden, J. D. W. and Baughman, R. H. [2007] "Polymer artificial muscles," *Materials Today* 10, 30–38.
- Moore, J. S., Bauer, J. M., Yu, Q., Liu, R. H., Devadoss, C., Jo, B.-H. and Beebe, D. J. [2000] "Functional hydrogel structures for autonomous flow control inside microfluidic channels," *Nature* 404, 588–590.
- Mora, T. and Boudaoud, A. [2006] "Buckling of swelling gels," The European Physical Journal E: Soft Matter 20, 119–124.
- Mullin, T., Deschanel, S., Bertoldi, K. and Boyce, M. C. [2007] "Pattern transformation triggered by deformation," *Physical Review Letters* 99, 084301.
- Na, J.-H., Evans, A. A., Bae, J., Chiappelli, M. C., Santangelo, C. D., Lang, R. J., Hull, T. C. and Hayward, R. C. [2015] "Programming reversibly self-folding origami with micropatterned photo-crosslinkable polymer trilayers," *Advanced Materials* 27, 79–85.
- Ng, T. Y., Li, H. and Yew, Y. K. [2010] "Computational analysis of smart soft hydrogels subjected to pH-electrical coupled stimuli: Effects of initial geometry," *International Journal of Solids and Structures* 47, 614–623.
- Oguz, O. [2007] Macroporous Hydrogels from Smart Polymers, Smart Polymers (CRC Press, Boca Raton, FL), pp. 269–297.

- Ohmine, I. and Tanaka, T. [1982] "Salt effects on the phase transition of ionic gels," The Journal of Chemical Physics 77, 5725–5729.
- Okumura, D., Inagaki, T. and Ohno, N. [2015] "Effect of prestrains on swelling-induced buckling patterns in gel films with a square lattice of holes," *International Journal of Solids and Structures* 58, 288–300.
- Okumura, D., Kuwayama, T. and Ohno, N. [2014] "Effect of geometrical imperfections on swelling-induced buckling patterns in gel films with a square lattice of holes," *International Journal of Solids and Structures* 51, 154–163.
- Oliveira, É. D., Silva, A. F. S. and Freitas, R. F. S. [2004] "Contributions to the thermodynamics of polymer hydrogel systems," *Polymer* 45, 1287–1293.
- Parson, W. W. [1978] "Thermodynamics of the primary reactions of photosynthesis," *Photochemistry and Photobiology* 28, 389–393.
- Peixinho, J. and Mukhopadhyay, S. [2013] "Diffusion-mechanical instability of a spherical gel," *Rencontre Du Non-linéaire*, 116–119.
- Peters, A. and Candau, S. J. [1986] "Kinetics of swelling of polyacrylamide gels," *Macro-molecules* 19, 1952–1955.
- Peters, A. and Candau, S. J. [1988] "Kinetics of swelling of spherical and cylindrical gels," *Macromolecules* 21, 2278–2282.
- Pritchard, R. H. and Terentjev, E. M. [2013] "Swelling and de-swelling of gels under external elastic deformation," *Polymer* 54, 6954–6960.
- Qiu, Y. and Park, K. [2001] "Environment-sensitive hydrogels for drug delivery," Advanced Drug Delivery Reviews 53, 321–339.
- Quesada-Perez, M., Ramos, J., Forcada, J. and Martin-Molina, A. [2012] "Computer simulations of thermo-sensitive microgels: Quantitative comparison with experimental swelling data," *The Journal of Chemical Physics* 136, 244903–244909.
- Richter, A., Paschew, G., Klatt, S., Lienig, J., Arndt, K.-F. and Adler, H.-J. P. [2008] "Review on hydrogel-based pH sensors and microsensors," Sensors 8, 561–581.
- Ricka, J. and Tanaka, T. [1984] "Swelling of ionic gels: Quantitative performance of the Donnan theory," *Macromolecules* 17, 2916–2921.
- Ryu, J., D'Amato, M., Cui, X., Long, K. N., Qi, H. J. and Dunn, M. L. [2012] "Photoorigami: Bending and folding polymers with light," *Applied Physics Letters* 100, 161908.
- Satarkar, N. S., Biswal, D. and Hilt, J. Z. [2010] "Hydrogel nanocomposites: A review of applications as remote controlled biomaterials," Soft Matter 6, 2364–2371.
- Satarkar, N. S. and Hilt, J. Z. [2008] "Magnetic hydrogel nanocomposites for remote controlled pulsatile drug release," *Journal of Controlled Release* 130, 246–251.
- Satarkar, N. S., Zhang, W., Eitel, R. E. and Hilt, J. Z. [2009] "Magnetic hydrogel nanocomposites as remote controlled microfluidic valves," *Lab on a Chip* 9, 1773–1779.
- Savin, T., Kurpios, N. A., Shyer, A. E., Florescu, P., Liang, H., Mahadevan, L. and Tabin, C. J. [2011] "On the growth and form of the gut," *Nature* 476, 57–62.
- Sheppard, Jr., N. F., Lesho, M. J., McNally, P. and Shaun Francomacaro, A. [1995] "Microfabricated conductimetric pH sensor," Sensors and Actuators B: Chemical 28, 95–102.
- Shimizu, S., Seyama, T., Sekine, R. and Kurita, K. [2003] "Effect of added organic solvent on interaction parameters of poly(N-isopropylacrylamide) in aqueous solution," *Journal of Applied Crystallography* 36, 694–697.
- Shirota, H., Endo, N. and Horie, K. [1998] "Volume phase transition of polymer gel in water and heavy water," *Chemical Physics* 238, 487–494.
- Siepmann, J., Kranz, H., Bodmeier, R. and Peppas, N. A. [1999] "HPMC-matrices for controlled drug delivery: A new model combining diffusion, swelling, and dissolution

mechanisms and predicting the release kinetics," *Pharmaceutical Research* **16**, 1748–1756.

- Sliozberg, Y. R., Chantawansri, T. L., Lenhart, J. L. and Andzelm, J. W. [2014] "Structural and mechanical properties of advanced polymer gels with rigid side-chains using coarsegrained molecular dynamics," *Polymer* 55, 5266–5275.
- Sugiura, S., Sumaru, K., Ohi, K., Hiroki, K., Takagi, T. and Kanamori, T. [2007] "Photoresponsive polymer gel microvalves controlled by local light irradiation," *Sensors and Actuators A: Physical* 140, 176–184.
- Sultan, E. and Boudaoud, A. [2008] "The buckling of a swollen thin gel layer bound to a compliant substrate," *Journal of Applied Mechanics* 75, 051002–051002.
- Sun, T. L., Kurokawa, T., Kuroda, S., Ihsan, A. B., Akasaki, T., Sato, K., Haque, M. A., Nakajima, T. and Gong, J. P. [2013] "Physical hydrogels composed of polyampholytes demonstrate high toughness and viscoelasticity," *Nature Materials* 12, 932–937.
- Suzuki, A. [1993] "Phase transition in gels of sub-millimeter size induced by interaction with stimuli," Dušek, K. (Ed.) (Springer, Berlin), pp. 199–240.
- Suzuki, A., Ishii, T. and Maruyama, Y. [1996] "Optical switching in polymer," Journal of Applied Physics 80, 131–136.
- Suzuki, A. and Kojima, S. [1994] "Phase transition in constrained polymer gel," The Journal of Chemical Physics 101, 10003–10007.
- Suzuki, A. and Suzuki, H. [1995] "Hysteretic behavior and irreversibility of polymer gels by pH change," *The Journal of Chemical Physics* **103**, 4706–4710.
- Suzuki, A. and Tanaka, T. [1990] "Phase transition in polymer gels induced by visible light," Nature 346, 345–347.
- Tanaka, H., Tomita, H., Takasu, A., Hayashi, T. and Nishi, T. [1992] "Morphological and kinetic evolution of surface patterns in gels during the swelling process: Evidence of dynamic pattern ordering," *Physical Review Letters* 68, 2794–2797.
- Tanaka, T., Fillmore, D., Sun, S.-T., Nishio, I., Swislow, G. and Shah, A. [1980] "Phase transitions in ionic gels," *Physical Review Letters* 45, 1636–1639.
- Tanaka, T. and Fillmore, D. J. [1979] "Kinetics of swelling of gels," The Journal of Chemical Physics 70, 1214–1218.
- Tanaka, T., Hocker, L. O. and Benedek, G. B. [1973] "Spectrum of light scattered from a viscoelastic gel," *The Journal of Chemical Physics* 59, 5151–5159.
- Tanaka, T., Sun, S.-T., Hirokawa, Y., Katayama, S., Kucera, J., Hirose, Y. and Amiya, T. [1987] "Mechanical instability of gels at the phase transition," *Nature* **325**, 796–798.
- Toh, W. [2015] Mechanics of soft matter: A modeling and simulation approach, School of Mechanical and Aerospace Engineering. Nanyang Technological University, Singapore.
- Toh, W., Ding, Z., Yong Ng, T. and Liu, Z. [2015] "Wrinkling of a polymeric gel during transient swelling," *Journal of Applied Mechanics* 82, 061004.
- Toh, W., Liu, Z. S., Ng, T. Y. and Hong, W. [2013] "Inhomogeneous large deformation kinetics of polymeric gels," *International Journal of Applied Mechanics* 5, 1350001.
- Toh, W., Ng, T. Y., Hu, J. and Liu, Z. [2014a] "Mechanics of inhomogeneous large deformation of photo-thermal sensitive hydrogels," *International Journal of Solids and Struc*tures 51, 4440–4451.
- Toh, W., Ng, T. Y., Liu, Z. S. and Hu, J. Y. [2014b] "The deformation kinetics of pHsensitive hydrogels," *Polymer International* 63, 1578–1583.
- Tomatsu, I., Peng, K. and Kros, A. [2011] "Photoresponsive hydrogels for biomedical applications," Advanced Drug Delivery Reviews 63, 1257–1266.
- Trujillo, V., Kim, J. and Hayward, R. C. [2008] "Creasing instability of surface-attached hydrogels," Soft Matter 4, 564–569.

- Uva, M., Pasqui, D., Mencuccini, L., Fedi, S. and Barbucci, R. [2014] "Influence of alternating and static magnetic fields on drug release from hybrid hydrogels containing magnetic nanoparticles," *Journal of Biomaterials and Nanobiotechnology* 5, 116–127.
- Valanis, K. C. and Landel, R. F. [1967] "The strain-energy function of a hyperelastic material in terms of the extension ratios," *Journal of Applied Physics* 38, 2997–3002.
- van der Linden, H. J., Herber, S., Olthuis, W. and Bergveld, P. [2003] "Stimulus-sensitive hydrogels and their applications in chemical (micro)analysis," Analyst 128, 325–331.
- Wallmersperger, T., Keller, K., Kröplin, B., Günther, M. and Gerlach, G. [2011]. Chemoelectro-mechanical modeling of pH-sensitive hydrogels, SPIE: Electroactive Polymer Actuators and Devices (EAPAD) p. 797610.
- Wallmersperger, T., Keller, K., Kröplin, B., Günther, M. and Gerlach, G. [2011b] "Modeling and simulation of pH-sensitive hydrogels," *Colloid Polymer Science* 289, 535–544.
- Wang, E., Desai, M. S. and Lee, S.-W. [2013] "Light-controlled graphene-elastin composite hydrogel actuators," *Nano Letters* 13, 2826–2830.
- Wang, J., Chen, Z., Mauk, M., Hong, K.-S., Li, M., Yang, S. and Bau, H. [2005] "Selfactuated, thermo-responsive hydrogel valves for lab on a chip," *Biomedical Microde*vices 7, 313–322.
- Wang, X. [2007] "Modeling the nonlinear large deformation kinetics of volume phase transition for the neutral thermosensitive hydrogels," *The Journal of Chemical Physics* 127, 1–8.
- Wang, X. and Hong, W. [2012] "A visco-poroelastic theory for polymeric gels," Proceedings of the Royal Society of London A: Mathematical, Physical and Engineering Sciences 471, 2182.
- Ward, M. A. and Georgiou, T. K. [2011] "Thermoresponsive polymers for biomedical applications," *Polymers* 3, 1215–1242.
- Weiss, F., Cai, S., Hu, Y., Kyoo Kang, M., Huang, R. and Suo, Z. [2013] "Creases and wrinkles on the surface of a swollen gel," *Journal of Applied Physics* **114**, 1–9.
- Wichterle, O. and Lim, D. [1960] "Hydrophilic gels for biological use," Nature 185, 117– 118.
- Wong, W. H., Guo, T. F., Zhang, Y. W. and Cheng, L. [2010] "Surface instability maps for soft materials," *Soft Matter* 6, 5743–5750.
- Wu, G., Xia, Y. and Yang, S. [2014] "Buckling, symmetry breaking, and cavitation in periodically micro-structured hydrogel membranes," *Soft Matter* 10, 1392–1399.
- Wu, Z., Bouklas, N. and Huang, R. [2013] "Swell-induced surface instability of hydrogel layers with material properties varying in thickness direction," *International Journal* of Solids and Structures 50, 578–587.
- Xiao, Z., Li, M. and Zhou, J. [2012] "Surface instability of a swollen cylinder hydrogel," Acta Mechanica Solida Sinica 25, 550–556.
- Yamaue, T. and Doi, M. [2004a] "Swelling dynamics of constrained thin-plate gels under an external force," *Physical Review E* 70, 11401.
- Yamaue, T. and Doi, M. [2004b] "Theory of one-dimensional swelling dynamics of polymer gels under mechanical constraint," *Physical Review E: Statistical, Nonlinear, and Soft Matter Physics* 69, 041402.
- Yamaue, T. and Doi, M. [2005] "The stress diffusion coupling in the swelling dynamics of cylindrical gels," *The Journal of Chemical Physics* **122**, 84703–84706.
- Yan, H., Jin, B., Gao, S. and Chen, L. [2014] "Equilibrium swelling and electrochemistry of polyampholytic pH-sensitive hydrogel," *International Journal of Solids and Structures* 51, 4149–4156.
- Yang, S., Khare, K. and Lin, P.-C. [2010] "Harnessing surface wrinkle patterns in soft matter," Advanced Functional Materials 20, 2550–2564.

- Yin, J., Bar-Kochba, E. and Chen, X. [2009] "Mechanical self-assembly fabrication of gears," Soft Matter 5, 3469–3474.
- Yin, J., Cao, Z., Li, C., Sheinman, I. and Chen, X. [2008] "Stress-driven buckling patterns in spheroidal core/shell structures," *Proceedings of the National Academy of Sciences* 105, 19132–19135.
- Yoon, J., Cai, S., Suo, Z. and Hayward, R. C. [2010] "Poroelastic swelling kinetics of thin hydrogel layers: Comparison of theory and experiment," Soft Matter 6, 6004–6012.
- Zalachas, N., Cai, S., Suo, Z. and Lapusta, Y. [2012] "Mechanical behavior of a pH-sensitive hydrogel ring used in a micro-optical device," PAMM 12, 411–412.
- Zalachas, N., Cai, S., Suo, Z. and Lapusta, Y. [2013] "Crease in a ring of a pH-sensitive hydrogel swelling under constraint," *International Journal of Solids and Structures* 50, 920–927.
- Zeng, X., Li, C., Zhu, D., Cho, H. J. and Jiang, H. [2010] "Tunable microlens arrays actuated by various thermo-responsive hydrogel structures," *Journal of Micromechanics* and Microengineering 20, 115035.
- Zhang, J., Zhao, X., Suo, Z. and Jiang, H. [2009] "A finite element method for transient analysis of concurrent large deformation and mass transport in gels," *Journal of Applied Physics* 105, 93522–93529.
- Zhang, X. X., Guo, T. F. and Zhang, Y. W. [2010] "Formation of gears through buckling multilayered film-hydrogel structures," *Thin Solid Films* 518, 6048–6051.
- Zhang, Y., Chen, L., Swaddiwudhipong, S. and Liu, Z. S. [2014] "Buckling deformation of annular plates describing natural forms," *International Journal of Structural Stability* and Dynamics 14, 1350054.
- Zhang, Y., Ji, H. F., Snow, D., Sterling, R. and Brown, G. M. [2004] "A pH sensor based on a microcantilever coated with intelligent hydrogel," *Instrumentation Science and Technology* 32, 361–369.
- Zhang, Y., Liu, Z. S., Swaddiwudhipong, S., Miao, H., Ding, Z. and Yang, Z. [2012] "pHsensitive hydrogel for micro-fluidic valve," *Journal of Functional Biomaterials* 3, 464– 479.
- Zhao, X., Hong, W. and Suo, Z. [2008] "Inhomogeneous and anisotropic equilibrium state of a swollen hydrogel containing a hard core," *Applied Physics Letters* 92, 051904.
- Zhao, X., Kim, J., Cezar, C. A., Huebsch, N., Lee, K., Bouhadir, K. and Mooney, D. J. [2010] "Active scaffolds for on-demand drug and cell delivery," *Proceedings of the National Academy of Sciences* 108, 67–72.
- Zhu, X., Wu, G., Dong, R., Chen, C.-M. and Yang, S. [2012] "Capillarity induced instability in responsive hydrogel membranes with periodic hole array," Soft Matter 8, 8088–8093.